Author's response to reviews

Title: A randomized controlled trial of a family-centred tobacco control program about environmental tobacco smoke to reduce respiratory illness in Indigenous infants.

Authors:

Vanessa Johnston (Vanessa.Johnston@menzies.edu.au)
Natalie Walker (n.walker@ctru.auckland.ac.nz)
David P Thomas (David.Thomas@menzies.edu.au)
Marewa Glover (m.glover@auckland.ac.nz)
Anne B Chang (Anne.Chang@menzies.edu.au)
Chris Bullen (c.bullen@ctru.auckland.ac.nz)
Peter Morris (Peter.Morris@menzies.edu.au)
Ngiare Brown (n.brown@usyd.edu.au)
Stephen Vander Hoorn (s.vanderhoorn@ctru.auckland.ac.nz)
Ron Borland (Ron.Borland@cancervic.org.au)
Catherine Segan (csegan@unimelb.edu.au)
Adrian Trenholme (atrenholme@middlemore.co.nz)
Toni Mason (Toni.Mason@cancervic.org.au)
Debra Fenton (Debra.Fenton@middlemore.co.nz)
Kane Ellis (kane.ellis@daniladilba.org.au)

Version: 3 Date: 25 February 2010

Author's response to reviews: see over
Author's response to reviews

Title: A randomized controlled trial of a family-centred tobacco control program about environmental tobacco smoke to reduce respiratory illness in Indigenous infants.

Authors:

Vanessa Johnston (Vanessa.Johnston@menzies.edu.au)
Natalie Walker (n.walker@ctru.auckland.ac.nz)
David P Thomas (David.Thomas@menzies.edu.au)
Marewa Glover (m.glover@auckland.ac.nz)
Anne B Chang (Anne.Chang@menzies.edu.au)
Chris Bullen (c.bullen@ctru.auckland.ac.nz)
Peter Morris (Peter.Morris@menzies.edu.au)
Ngiare Brown (n.brown@usyd.edu.au)
Stephen Vander Hoorn (s.vanderhoorn@ctru.auckland.ac.nz)
Ron Borland (Ron.Borland@cancervic.org.au)
Catherine Segan (c.segan@unimelb.edu.au)
Adrian Trenholme (atrenholme@middlemore.co.nz)
Toni Mason (Toni.Mason@cancervic.org.au)
Debra Fenton (Debra.Fenton@middlemore.co.nz)
Kane Ellis (kane.ellis@daniladilba.org.au)

Version: 2 Date: 25 February 2010

Author's response to reviews: see over
Editor,  
BMC Public Health  

25 February 2010  

Dear Editor,  

Please find attached the protocol for a new randomized controlled trial - A randomized controlled trial of a family-centred tobacco control program about environmental tobacco smoke to reduce respiratory illness in Indigenous infants.  

I have made the suggested revision to this paper, namely I have inserted a statement (in tracked changes) on p.13 that states “documented written consent will be obtained from all participants prior to entering the study.”  

I have also moved the ‘competing interests’ section to before the ‘authors’ contributions’ in keeping with the journal’s style layout.  

Yours sincerely,  

[Signature]  

Dr. Vanessa Johnston  
Menzies School of Health Research  
PO Box 41096  
Casuarina  
NT 0811  

pH: 61 8 8922 7968  
e: Vanessa.johnston@menzies.edu.au